Title: Family doctors’ problems and motivating factors in management of depression

Authors:

Pille Oopik (pille.oopik@ut.ee)
Anu Aluoja (anu.aluoja@klinikum.ee)
Ruth Kalda (ruth.kalda@ut.ee)
Heidi-Ingrid Maaroos (heidi-ingrid.maroos@ut.ee)

Version: 7 Date: 15 September 2006

Author’s response to reviews: see over
Authors` response to reviewers.

Title: Family doctors’ problems and motivating factors in management of depression
Authors:
Pille Ööpik: pille.oopik@ut.ee
Anu Aaluja: anu.aluoja@kliinikum.ee
Ruth Kalda: ruth.kalda@ut.ee
Heidi-Ingrid Maaroos: heidi-ingrid.maaroos@ut.ee

14.09.2006

We would like to re-submit for your consideration our revised manuscript entitled: “Family doctors' problems and motivating factors in management of depression” by Pille Ööpik, Anu Aaluja, Ruth Kalda, Heidi-Ingrid Maaroos.

Our article has been linguistically revised.